# Land-use change contributing almost half of future diversity change of global terrestrial vertebrates under climate change

Xiaojuan Liu<sup>1</sup>, Xia Li<sup>1</sup>, Jinbao Zhang<sup>2</sup>, Han Zhang<sup>2</sup>, Hong Shi<sup>3</sup>, Yuchao Yan<sup>4</sup>, and Han Zhang<sup>1</sup>

<sup>1</sup>East China Normal University <sup>2</sup>Sun Yat-sen University <sup>3</sup>Southwest Minzu university <sup>4</sup>Peking University

November 24, 2022

#### Abstract

Global biodiversity is lost at an unprecedented ratio driven by climate change and land-use change. However, little is known about the combined effects of climate and land-use change on future biodiversity on a global scale. Here, we first build the indices of land-use naturalness and the land-use intensity to depict the land-use change on a global scale. By using random forest models, we establish the empirical relationship to quantify this combined effect and further predict future changes of terrestrial vertebrates can be predicated under the Shared Socio-economic Pathways (SSPs). The relative contributions of climate change and land-use change to terrestrial vertebrates are finally separated through quantitative analysis. We find that future land-use change contributes to 48.52% of richness changes, slightly lower than that of climate change. Nearly 45.82% of the Earth's land will suffer richness losses of terrestrial vertebrates by 2050 even under the middle-high scenario of SSP3, mainly located at low latitudes, such as Southeast Asia, Latin America and sub-Saharan Africa. Moreover, the analysis at the country-specific level reveals that nearly half of the world's countries would experience species richness loss in the nearby future. These findings make clear that both climate change and land-use change pose comparably significant threats to global biodiversity. More immediate attention and effective actions are urgently needed from local governments for vulnerable regions.

1	L	and-use change contributing almost half of future diversity change
2		of global terrestrial vertebrates under climate change
3	Xia	aojuan Liu <sup>a</sup> , Xia Li <sup>a*</sup> , Jinbao Zhang <sup>b</sup> , Han Zhang <sup>b</sup> , Hong Shi <sup>c</sup> , Yuchao Yan <sup>d</sup> , Han Zhang <sup>a</sup>
4	a.	Key Lab. of Geographic Information Science (Ministry of Education), School of Geographic
5		Sciences, East China Normal University, Shanghai 200241, P.R. China
6	b.	School of Geography and Planning, Guangdong Key Laboratory for Urbanization and
7		Geo-simulation, Sun Yat-sen University, Guangzhou 510275, China.
8	c.	School of Tourism and Historical Culture, Southwest Minzu university, Chengdu 610064, P.R.
9		China
10	d.	Sino-French Institute for Earth System Science, College of Urban and Environmental Sciences,
11		Peking University, Beijing 100871, P.R. China
12	*C	orresponding authors: Xia Li (lixia@geo.ecnu.edu.cn)
13		Key Lab of Geographic Information Science (Ministry of Education),

- 14 School of Geographic Sciences,
- 15 East China Normal University,
- 16 500 Dongchuan Rd., Shanghai 200241, P.R. China.

#### 17 Abstract

18 Global biodiversity is lost at an unprecedented ratio driven by climate change and 19 land-use change. However, little is known about the combined effects of climate and 20 land-use change on future biodiversity on a global scale. Here, we first build the 21 indices of land-use naturalness and the land-use intensity to depict the land-use 22 change on a global scale. By using random forest models, we establish the empirical 23 relationship to quantify this combined effect and further predict future changes of 24 terrestrial vertebrates can be predicated under the Shared Socio-economic Pathways 25 (SSPs). The relative contributions of climate change and land-use change to terrestrial 26 vertebrates are finally separated through quantitative analysis. We find that future 27 land-use change contributes to 48.52% of richness changes, slightly lower than that of 28 climate change. Nearly 45.82% of the Earth's land will suffer richness losses of 29 terrestrial vertebrates by 2050 even under the middle-high scenario of SSP3, mainly 30 located at low latitudes, such as Southeast Asia, Latin America and sub-Saharan 31 Africa. Moreover, the analysis at the country-specific level reveals that nearly half of 32 the world's countries would experience species richness loss in the nearby future. 33 These findings make clear that both climate change and land-use change pose comparably significant threats to global biodiversity. More immediate attention and 34 35 effective actions are urgently needed from local governments for vulnerable regions. 36 **Keywords**: biodiversity loss; richness changes; climate change; land-use change; 37 terrestrial vertebrates; Shared Socio-economic Pathways (SSPs)

2 / 41

#### 38 Plain Language Summary

39 With the increasing rate of climate change and human pressure on land, mitigating the 40 loss of global biodiversity is a major challenge for the world's organizations and 41 nations. In this study, we assess the combined effect of climate change and land-use 42 change on diversity changes of global terrestrial vertebrates under the shared 43 socioeconomic pathways (SSPs) and evaluate the relative contributions of climate 44 change and land-use change to these changes. We find that approximately 45.82% of 45 the Earth's land will suffer the richness loss of terrestrial vertebrates by 2050 even 46 under the middle-high scenario of SSP3. All the projections of these five SSPs 47 scenarios show that species richness changes of terrestrial vertebrates have unique geographical variations. Low latitudes  $(20^{\circ}\text{S} - 25^{\circ}\text{N})$  will experience a sharp decline 48 49 in species richness, while high latitudes (> 60°N) will experience a slight increase. 50 Moreover, nearly half of the world's countries would experience diversity loss in the 51 nearby future. These changes are predicted to contribute more climate change 52 (accounting for 51.48%) than land-use change (nearly 48.52%) at a global scale under 53 SSPs, which indicates that global land-use change plays a comparable role, compared 54 with climate change, in future biodiversity.

## 55 **1. Introduction**

56 Global biodiversity plays an important role in ecosystem functions, as well as in 57 the development of human well-being (Xu et al., 2021). Biodiversity is strongly associated with the productivity and resilience of terrestrial ecosystems through 58 59 changing the rate of decomposition (Balvanera et al., 2006), carbon cycle (Midgley, 60 2012) and interspecies relationships (Wardle, Bardgett, Callaway, & Van der Putten, 61 2011). Moreover, it is also closely related to products, such as food supply and 62 pharmaceutical products, that are essential in human life by mediating pollination and 63 other processes (Booth et al., 2021). Nevertheless, global biodiversity has experienced an increasing loss since the Anthropocene (Johnson et al., 2017). The "Global Risk 64 Report 2020" published by the World Economic Forum (WEF) also ranked 65 "biodiversity loss" as the second most impactful and third most likely risk for the next 66 67 decade. Factors driving biodiversity loss are widely varied, ranging from natural 68 processes to anthropogenic activities (Maxwell, Fuller, Brooks, & Watson, 2016). 69 Many studies attribute the biodiversity loss to climate change (Di Marco et al., 2019; 70 Hickling, Roy, Hill, Fox, & Thomas, 2006; Mantyka-pringle, Martin, & Rhodes, 71 2012). Land-use change also poses a serious threat to global biodiversity. However, we, at present, cannot fully understand the combined effect of climate and land-use 72 73 change on biodiversity loss at a global scale.



Climate change is considered as a primary factor driving biodiversity loss.

4 / 41

75 Continuous rising of temperature can directly change the natural environment of habitats, which eventually leads to a widespread species extinction. Recent studies 76 77 have shown that species exhibit several responses to climate change. For instance, 78 Parmesan (2006) found that evolutionary adaptations to warmer conditions were 79 important for species against climate change. Poleward shifts of species' ranges 80 (Hickling et al., 2006) and species invasion (Dornelas et al., 2014) are also common 81 responses to climate change. However, not all species can shelter themselves from the 82 negative effects of climate change by adaptation or migration. Some studies indicated 83 that range-restricted species, like species ranged in polar or mountaintop, are more 84 likely to undergo extinct (Dullinger et al., 2012).

85 Land-use change can increase the risk of species extinction combined with climate change by exacerbating the removal and fragmentation of native habitat in 86 87 some regions. Peters et al. (2019) suggested that land-use change in climate-sensitive 88 areas is likely to amplify the negative effect caused by climate change. For example, 89 the land-use change in arid and semi-arid lands which is sensitive to climate 90 conditions can increase the risk of species richness loss (Davies et al., 2012). 91 Similarly, the agricultural expansion and urban sprawl aggravate the richness loss of 92 soil organic carbon caused by climate change in wetland areas (Rojas, Munizaga, 93 Rojas, Martínez, & Pino, 2019). On the other hand, the negative effects of climate 94 change on biodiversity can also be ameliorated by land-use change. More recently, 95 studies noticed that building protected areas could effectively resist the negative 5/41

97 2020). Besides, land-use conversions with less attention paid on, such as from 98 agricultural land to forests, can also offset part of the negative effects from climate 99 change (Manaye, Negash, & Alebachew, 2019). In addition, the magnitude of 100 land-use intensity varies markedly at a global scale may cause varied consequences to 101 different biodiversity changes (Pekin & Pijanowski, 2012).

effects of climate change (Maiorano, Falcucci, Garton, & BOITANI, 2007; Shi et al.,

96

102 To comprehensively reveal the combined effect on biodiversity changes, a solid 103 understanding of the potential effects of future change on biodiversity as well existing 104 status is necessary. Scenario-based biodiversity projection is essential for predicting 105 the potential biodiversity loss. Scenario-based biodiversity projection should be 106 essential for predicting the potential biodiversity loss and measuring the effectiveness 107 of protection measures. Future scenarios, in general, should incorporate 108 social-economic factors, such as human population density, economic development 109 and greenhouse gases emissions. This requirement can be addressed by the recently 110 generated scenario, the Shared Socio-economic Pathways (SSPs) (Riahi et al., 2017). 111 However, limited studies have been carried out to quantify the combined effect of 112 climate and land-use change on biodiversity change. It is unclear about which factors 113 may dominate the biodiversity change on the global scale and what is the spatial 114 heterogenicity of their influences.

115 Here we attempt to quantify the combined effect of climate and land-use change 116 on diversity changes of terrestrial vertebrates under SSPs, and explore the relative  $\frac{6}{41}$  117 contributions of climate change and land-use change to these changes. To be specific, 118 we aim to answer: (a) How the combined effect of climate and land-use change on 119 diversity changes of terrestrial vertebrates at the global scale? (b) Compared with 120 climate change, how much does future land-use change contribute to diversity 121 changes of terrestrial vertebrates?

## 122 **2. Materials and methods**

123 This section provides a summary of dataset collection and some methods used in 124 this paper. First, the land-use naturalness and land-use intensity proxies on a global 125 scale were generated for global land-use change with land-use data, net primary productivity (NPP) and population density. Second, we built the species distribution 126 127 models with climatic and land-use variables by using empirical data. Two methods, generalized additive models (GAM) and random forest (RF) methods, were then 128 evaluated to choose the best with a higher value of  $R^2$  for the prediction. Third, the 129 130 combined effect of climate and land-use change on terrestrial vertebrates was assessed 131 by projecting the species richness changes under SSPs. Finally, we evaluated the 132 relative contributions of climate and land-use change for future biodiversity change 133 which may depend critically on the land-use change on the global scale.

134 **2.1 Diversity measures** 

135 Species richness, which measures the number of different species in an 136 ecological sample, is a biodiversity index that formed the basis for various 7/41 biodiversity studies (Jenkins, Pimm, & Joppa, 2013). The species richness was also
adopted as a proxy of global diversity of terrestrial vertebrates in this paper. For
amphibians and mammals, we employed the geographical distribution database from
the International Union for the Conservation of Nature (IUCN) Red List
(https://www.iucnredlist.org/). As for birds, we used the species distribution data from
the Birdlife International (http://www.birdlife.org/).

To generate richness maps on a global scale, we first removed the species range polygons which were classified as "extinct", "extinct in the wild", "not evaluated" and "data deficient" categories, and unionized polygons with the same taxonomic name. We then created a fishnet with a spatial grain of 1km×1km by using ArcGIS, and counted the overlap between species range polygons in each grid cell. The final generated world's richness maps of terrestrial vertebrates involved 4,708 mammal species, 5,208 amphibian species, and 17,228 bird species.

150 **2.2 Climate and land-use variables** 

To incorporate climatic variables and elevation into our analysis, we considered the following climate variables: mean annual temperature (Fadrique et al., 2018), mean annual precipitation (Garcia, Cabeza, Rahbek, & Araújo, 2014), mean annual wind speed (Porter, Budaraju, Stewart, & Ramankutty, 2015) and mean elevation (Elsen & Tingley, 2015). The mean elevation was chosen to reflect the effect of altitude on species richness when building species distribution models. Climate 157 variables were derived from the Global Surface Summary of the Day (https://data.noaa.gov/dataset/global-surface-summary-of-the-day-gsod) 158 and the Coupled Model Intercomparison Project Phase 6 (CMIP 6, https://esgf-node.llnl.gov/). 159 160 The dataset of the global Surface Summary of the Day and the CMIP6 were used for 161 building RF models and predicting the future geographic distribution of terrestrial 162 vertebrates, respectively. Here, we chose the climate variables from the CMIP6 with 163 the combination of SSP1-RCP1.9, SSP2-RCP4.5, SSP3-RCP7.0, SSP4-RCP6.0 and SSP5-RCP8.5 for 2050. We resampled these four climate variables and the mean 164 165 elevation into 30-arc resolution (http://www.fao.org/), and excluded the Antarctic area and the grid cells with missing climate information. 166

As the species richness of terrestrial vertebrates is also sensitive to land-use change (Newbold et al., 2016). We defined land-use naturalness (LUN) and the land-use intensity (LUI) as two proxies to detect the land-use change on a global scale. As shown in equation (1), the LUN was described as the product of the average naturalness (*Anat*) and net primary productivity (NPP, from http://files.ntsg.umt.edu/). The LUI was related to *Anat* and population density of human being (POP, from https://landscan.ornl.gov/landscan-datasets).

174 
$$LUN = Anat \times NPP$$
 (1)

175 
$$LUI = (1 - Anat) \times POP$$
(2)

176 where, *Anat* was associated with the land-use categories and values of naturalness.

177 To calculate the LUN and LUI, the Intergovernmental Panel on Climate Change 9/41 178 land categories and the European Space Agency Climate Change Initiative Land 179 Cover (CCI-LC) land dataset (http://maps.elie.ucl.ac.be/CCI/viewer/index.php), with 180 300-meter spatial resolution in 2017 and its corresponding land categories were 181 adopted. According to the correspondence of the two categories (Table 1), the CCI-LC 182 land-use classes were grouped into the six IPCC land categories, for instance, 183 agriculture, forest, grassland, wetland, settlement and other land. The value of 184 naturalness in each land-use class was referred from Montesino et al.(2014). We 185 further calculated the values *Anat* for five land use classes according to Equation (3).

$$Anat = \frac{\sum_{i=0}^{k} nat_i \times ncell_i}{\sum_{i=0}^{k} ncell_i}$$
(3)

187 In which,  $nat_i$  and  $ncell_i$  are the value of naturalness and the number of grid cells 188 in the *i*th land cover class, respectively. *k* is the number of land cover classes.

186

For calculating the two land-use proxies under SSPs, we also employed the projected land use data, NPP and world population density from the Integrated Model to assess the Global Environment (IMAGE, https://dataplatform.knmi.nl/?q=PBL) (Popp et al., 2017). As the projected land-use data were cover percentages of different land-use classes in each grid cell, the detailed calculation of land-use naturalness and land-use intensity was according to Equation (4) and Equation (5).

195 
$$LUN_{SSP} = \left(\sum_{i=0}^{k} Anat_{i}\right) \times NPP_{SSP}$$
(4)

196 
$$LUI_{SSP} = \left(\sum_{i=0}^{k} (1 - Anat_i)\right) \times POP_{SSP}$$
(5)

197 In which,  $LUN_{SSP}$  and  $LUI_{SSP}$  were the LUN and LUI under SSPs. NPP<sub>SSP</sub> and 198 POP<sub>SSP</sub> were the net primary productivity and population density under SSPs. *k* was 10/41 199 the number of land cover classes.

200 
$$cell_j = \frac{Vcell_j - raster_{min}}{raster_{max} - raster_{min}} (j = 1, 2, ..., n)$$
(6)

All climate and land-use variables in our analysis were normalized according to Equation (6). In which,  $Vcell_j$  was the original value in the *j*th grid cell,  $raster_{min}$ and  $raster_{max}$  were the minimum and maximum values in raster data, respectively. *n* indicated the number of grid cells in raster data.

Table 1 The correspondence b	between the land cate	gories and values of	of naturalness in eac	ch land classes
------------------------------	-----------------------	----------------------	-----------------------	-----------------

Land categories (IPCC)	Average naturalness	Land categories (CCI-LC)	Naturalness
Agriculture	0.22	Rained cropland	0.20
		Irrigated cropland	0.25
		Mosaic cropland (>50%) / natural vegetation (tree, shrub, herbaceous cover) (<50%)	0.30
		Mosaic natural vegetation (tree, shrub, herbaceous cover)	0.90
		(>50%) / cropland (< 50%)	
Forest	0.87	Tree cover, broadleaved, evergreen, closed to open (>15%)	0.95
		Tree cover, broadleaved, deciduous, closed to open (> 15%)	0.90
		Tree cover, needleleaved, evergreen, closed to open (> 15%) $)$	0.90
		Tree cover, needleleaved, deciduous, closed to open (> 15%)	0.85
		Tree cover, mixed leaf type (broadleaved and needleleaved)	0.70
		Mosaic tree and shrub (>50%) / herbaceous cover (< 50%)	0.60
		Tree cover, flooded, fresh or brakish water	0.50
		Tree cover, flooded, saline water	0.45
Grassland	0.77	Mosaic herbaceous cover (>50%) / tree and shrub (<50%)	0.40
		Grassland	0.80
Wetland	0.85	Shrub or herbaceous cover, flooded, fresh-saline or brakish	0.85
		water	
Settlement	0.00	Urban	0.00
Other	0.17	Shrubland	0.30
		Lichens and mosses	0.15
		Sparse vegetation (tree, shrub, herbaceous cover)	0.20
		Bare areas	0.10

## 206 **2.3 Statistical analysis**

The species distribution model (SDM) is commonly used for predicting the 207 208 geographical distribution of species and providing some evidence for species endangerment assessment. The SDM assumes that the niche for each specie depends 209 210 on the environmental factors in its habitat. According to recent studies (Barbet-Massin, 211 Thuiller, & Jiguet, 2012; Thuiller, Lafourcade, Engler, & Araújo, 2009), generalized 212 additive models (GAM) and machine learning algorithms are more widely used for 213 solving the geographic distribution of species. For example, Montesino et al.(2014) 214 adopted GAM to assess the effect of future land-use change on biodiversity in global protected areas. Marmion et al.(2009) compared eight modelling techniques for 215 216 predicting plant geographical distribution in North-eastern Finland and found that RF method performed the best. However, it remains unclear which one performs better on 217 218 a global level.

Therefore, we compared GAM with RF for accessing the potential effects of climate and land-use change on species richness of terrestrial vertebrates, as well as that of different taxa. We parameterized GAM by default settings with the pyGAM package in Python 3.6. For random forest methods, the number of trees and the maximum number of features were set to be 100 and 6, respectively. To evaluate model performance, we split the dataset into training and testing sets through 10-fold cross-validation and calculated the adjusted  $R^2$ . The Terrestrial Ecoregions of the World (TEOW) data was also introduced to improve the accuracy of the species distribution model. The TEOW data was derived from the World Wildlife Fund and defined 867 terrestrial ecoregions that classified into 14 biomes across the globe (Olson et al., 2001). For each biome, we carried out species distribution models and selected the model with a higher adjusted  $R^2$  from GAM and RF.

## 231 **2.4 Contributions of climate and land-use factors to richness changes**

232 To quantify the contributions of climate change and land-use change to richness 233 changes of terrestrial vertebrates, we predicted the spatially land-use-induced and 234 climate-driven distribution of species richness under the SSPs with RF models respectively. Specially, the spatially land-use-induced species richness (Bioland) was 235 simulated by using the constant land-use dataset and future climate dataset. Similarly, 236 the climate-driven species richness (Bioclimate) was projected with the constant 237 238 climate dataset and future land-use dataset. Here, we considered the projected species richness with future climate and land-use dataset as the actual species richness under 239 240 SSPs. Accordingly, we could calculate the difference between the actual species 241 richness and the Bioland, and the difference between the actual species richness and 242 *Bio<sub>climate</sub>* using Equation (7) and Equation (8).

$$\Delta Bio_{land} = Bio_{land} - Bio_{land\_climate}$$
(7)

$$\Delta Bio_{\text{climate}} = Bio_{\text{climate}} - Bio_{\text{land_climate}} \tag{8}$$

245 Following Wu et al.(2014) and Liu et al.(2019), we estimated the relative

14 / 41

contributions of climate change and land-use change to richness changes of terrestrial vertebrates as Equation (9) and Equation (10). Here, we considered the sum of contributions of climate change and land-use change to be 100%. The final contributions of climate change and land-use change to the loss under different SSPs were processed according to the terrestrial biomes using the zonal statistics of ArcGIS.

252 
$$\operatorname{Contr}_{land} = \frac{|\Delta Bio_{land}|}{|\Delta Bio_{land}| + |\Delta Bio_{climate}|} \times 100\%$$
(9)

253 
$$\operatorname{Contr}_{climate} = \frac{|\Delta Bio_{climate}|}{|\Delta Bio_{land}| + |\Delta Bio_{climate}|} \times 100\%$$
(10)

## 254 **3. Results**

#### **3.1 Performance of species distribution models**

256 After determining the correlation between species richness and climate and 257 land-use variables with Pearson correlation analysis (SI. Figure 1), we used random 258 forest (RF) models to build species distribution models for global terrestrial 259 vertebrates, and compared the results with those from a generalised additive model (GAM). The results show that models including climate and land-use variables have 260 261 higher explanatory power for species distribution than models that only use climate variables (SI. Table 1). Additionally, the RF methods show generally higher 262 performance than GAM for terrestrial vertebrates, as well as for amphibians, 263 mammals and birds (Figure 1). Specifically, the RF methods have higher explanatory 264 265 power for the species richness of terrestrial vertebrates in Tropical and Subtropical

266 Coniferous Forests (TSC, abbreviations of all terrestrial biomes can be referred to SI. Table 2) than GAM ( $R^2 = 0.79$  for RF and  $R^2 = 0.55$  for GAM). Similarly, the species 267 richness of amphibians, mammals and birds are also can strongly explained by using 268 random forest methods, but moderately explained by GAM, especially in TSC and 269 MWS (with  $R^2 = 0.70$ , 0.81 and 0.72 for RF and with  $R^2 = 0.49$ , 0.53 and 0.59 for 270 271 GAM). Even under the poorest situation, the RF method still behaves fairly with the 272 GAM. For example, the species richness of mammals in BRF explains by RF method with  $R^2 = 0.62$  and by the GAM with  $R^2 = 0.63$ , which is slightly higher than random 273 274 forest method (but less than 0.01). Therefore, we choose the RF methods to predict the species richness of terrestrial vertebrates under SSPs. 275



276

277 Figure 1 R<sup>2</sup> of generalized additive models (GAM) and random forest (RF) methods. Response 278 variables in a-d are the species richness of terrestrial vertebrates, amphibians, mammals and birds, 279 respectively. The red dash line represents the value of  $R^2$  equals 0.7. TMB, TDB, and TSC denote 280 the biomes of Tropical and Subtropical Moist Broadleaf Forests, Tropical and Subtropical Dry 281 Broadleaf Forests, and Tropical and Subtropical Coniferous Forests. TBM, TCF and BRF 282 represent the biomes of Temperate Broadleaf and Mixed Forests, Temperate Coniferous Forests, 283 and Boreal Forests/Taiga. TSG, TGS, and FGS are the biomes of Tropical and Subtropical 284 Grasslands, Savannas, and Shrublands, Temperate Grasslands, Savannas, and Shrublands, and 285 Flooded Grasslands and Savannas. MGS, TDA, and MWS are the biomes of Montane Grasslands 286 and Shrublands, Tundra, and Mediterranean Forests, Woodlands, and Scrub. DXS and MGV are 287 the biomes of Deserts and Xeric Shrublands and Mangroves, respectively. The abbreviation of 14 288 terrestrial biomes also can be referred to SI. Table 2.

289 **3.2 Projected richness changes under SSPs** 

290	We predicted the changes of species richness for terrestrial vertebrates across the
291	globe under SSPs with the combined effects of climate and land-use change. The
292	estimation shows that about 45.99% of the world's land would suffer a loss of species $17/41$

293 richness between 2017 and 2050 under climate and land-use change. The magnitude and geographic distribution of the changes vary under the five different SSPs (SI. 294 295 Figure 3 - 6). In general, the heaviest richness loss is projected under SSP5 (with 296 46.29% of terrestrial land suffering richness loss), but the lowest species richness 297 under SSP3, with about 45.82% of global land experiencing richness loss (Table 2). 298 To be specific, the differences between the five SSPs are mainly distributed in Latin 299 America and Southeast Asia. For instance, the magnitude of richness loss in the Guiana Highlands is largest under SSP5, followed by that under SSP2. In contrast, the 300 301 loss of species richness under SSP3 is estimated the least compared with the other 302 four SSPs, no matter in magnitudes or geographical ranges. As shown in Figure 2, 303 Southeast Asia will suffer the most significant richness loss of terrestrial vertebrates, 304 with a maximum loss of 305 species (nearly 83.33% of the present species richness) 305 in the Malay Archipelago by 2050 under SSP3. These richness losses are close to the 306 results of Chaudhary and Mooers (Chaudhary & Mooers, 2018), who predicted a loss 307 of nearly 281 species under land-use change from 2050 to 2100. Interestingly, the 308 richness loss in Latin America is concentrated in the east of the Brazilian plateau and 309 the North Cordillera Mountains but scattered around the Amazon Basin. In terms of 310 quantity, the richness loss in Latin America is slightly lower than that in Southeast 311 Asia, with a maximum richness loss of 187 species.

SSPs	Terrestrial vertebrates	Amphibians	Mammals	Birds
SSP1	45.98	40.90	42.94	38.92
SSP2	46.00	40.52	42.62	38.94
SSP3	45.82	40.75	42.82	38.61
SSP4	45.85	40.21	42.37	38.75
SSP5	46.29	41.13	43.07	39.16

 Table 2 The percentages of terrestrial land with richness loss under SSPs (%)

313 For different taxa, we find that mammals have the largest geographical range size 314 with richness loss by 2050 (about 42.76% of the world's land), followed by amphibians (about 40.70% of the world's land). Although mammals show the largest 315 316 geographical range size with richness loss, the quantity of richness loss is far less than 317 that of other taxa. To be specific, the heaviest richness loss of amphibians is estimated 318 to be 84 species, while that of mammals is 65 species. Furthermore, the richness 319 changes for different taxa shows geographical variation. The richness loss of 320 amphibians is mainly distributed in the Amazon Basin and the Brazilian Plateau in 321 Latin America, south Congo Basin and the Atlantic Coastal Plain, whereas the 322 richness increase of amphibians is distributed in the north Amazon, the Congo Basin, 323 the Yunnan-Guizhou Plateau, and Papua Islands. For birds, the richness increase 324 under SSP3 is mainly located in the Congo Basin, Papua Islands and the 325 Yunnan-Guizhou Plateau.

a The rate of species richness changes



b The number of species richness changes



326

Figure 2 Richness changes of terrestrial vertebrates between 2017 and 2050 under
SSP3. a. the rate of richness changes (%), b. the number of richness changes.

In addition, richness changes at low latitudes  $(20^{\circ}S - 25^{\circ}N)$  and low elevation (< 330 1500 meter) are projected to decline sharply. As shown in Figure 3, the richness losses 331 of terrestrial vertebrates, as well as amphibians, mammals, and birds, are mainly 332 distributed at latitudes between 20°S and 25°N under the SSPs and are projected to 333 experience a large fluctuation in magnitude. For instance, the species richness of

334	terrestrial vertebrates will decline by nine species per year but increase by two species
335	per year from the present to 2050. In contrast, the magnitude of richness changes
336	around the 60° magnetic latitude and higher is relatively small, showing a slightly
337	increasing trend. These results show a whole range shift from low latitude to high
338	latitude as a result of climate and land-use change, coincident with those of previous
339	studies (Chen, Hill, Ohlemüller, Roy, & Thomas, 2011; Hill, Griffiths, & Thomas,
340	2011; Pauli et al., 2012). The comparison between different taxa emphasizes that
341	mammals are more likely to suffer richness loss in the middle of the 21th Century, no
342	matter at low latitudes (Figure 3b, c, d). Furthermore, our projections find the loss of
343	species richness is concentrated at low altitudes (< 1500m). Taking SSP3 for example
344	(SI. Figure 2), the largest loss of species richness below 1500m reached five species
345	per year, with birds experiencing the largest loss, followed by mammals.



346

Figure 3 Annual changes of species richness along with magnetic latitudes
between 2017 and 2050 under SSP3 for a. terrestrial vertebrates, b. amphibians, c.
mammals and d. birds. Colour bar shows the number of grid cells that located in annual
change of richness species and latitude.

351	The projection also indicates that nearly half of the world's countries would
352	experience a richness loss by 2050. In general, approximately 19.62% of world's
353	countries have an average rate of species-richness loss over 30.00%, and 17.72% of
354	countries have an average rate of increase over 30.00%. By introducing the Human
355	Development Index (HDI), the numbers of high-income countries with richness loss
356	and increase are almost equal, but the rate of increase is larger than the rate of loss
357	(Figure 4b). Similarly, for low-income countries, the number of countries with
358	richness loss is also equivalent to that with richness increase. However, compared

with countries at the high-income level, the magnitudes of species-richness changes
are much slighter. It is worth mentioning that the largest rate of species-richness loss
(74.17%) is estimated at middle-income countries.





363 Figure 4 The average of net richness changes (%) for terrestrial vertebrates at the country level. a. the average richness loss of terrestrial vertebrates for the top 50 of the 364 world's countries between 2017 and 2050 under SSPs. Orange solid lines indicate the 365 366 95% confidence interval of country-specific richness loss. Labels in vertical axis are the country code (ISO3) of the top 50 world countries. b. Country-level richness changes 367 between 2017 and 2025 in relation to countries' human development index (HDI). 368 369 Colours represent the countries in different income levels and the point diameter 370 indicates the value of species-richness loss.

371 **3.3 Contributions of climate change and land-use change** 

372	We estimated	the relative	contributions	of future	climate	change	and	land-use
-----	--------------	--------------	---------------	-----------	---------	--------	-----	----------

373 change to species richness changes. The analysis reveals that the contribution of 23/41

land-use change to biodiversity changes can reach 48.52% on average, which is even
slightly lower than that of climate change (51.48%) on the global scale. However, the
relative contributions of climate change and land-use change are varied among the
five scenarios of SSPs. For instance, the climate-related contribution is the largest
under SSP5 (accounting for 51.29%) but lowest under SSP3 (estimated at 50.45%).

379 Contributions of climate change and land-use change show obvious variations in 380 terrestrial biomes. Taking SSP5 as an example, climate change shows the greatest 381 effect on richness changes in FGS (68.78%), followed by that in TSC (66.05%), 382 whereas land-use change has the strongest influence in BRF (68.72%). As table 3 383 shows, the biomes of TSC, TGS and FGS are highly explained by climate change 384 under the five SSPs (with climate-related contribution beyond 60%), which indicates 385 that terrestrial vertebrates in these biomes are more vulnerable to climate change than 386 land-use change. Instead, the biomes of BRF and TDA are prone to be influenced by 387 land-use change under all the SSPs, with land-use-related contributions at 69.43% and 388 60.88% in SSP3, respectively.

	SS	SP1	SS	SP 2	SS	SP 3	SS	SP 4	SS	SP 5
Biomes	climate	land-use								
	change	change								
TMB	45.32	54.68	46.50	53.50	46.46	53.54	45.91	54.09	46.33	53.67
TDB	46.86	53.14	47.44	52.56	49.35	50.65	47.14	52.86	51.10	48.90
TSC	60.42	39.58	64.01	35.99	63.78	36.22	61.15	38.85	66.05	33.95
TBM	48.40	51.60	47.31	52.69	48.10	51.90	45.59	54.41	47.95	52.05
TCF	54.25	45.75	51.60	48.40	53.30	46.70	52.88	47.12	52.19	47.81
BRF	31.44	68.56	30.37	69.63	30.57	69.43	30.65	69.35	31.28	68.72
TSG	45.84	54.16	47.04	52.96	47.64	52.36	46.36	53.64	47.36	52.64
TGS	65.98	34.02	63.99	36.01	65.72	34.28	65.45	34.55	64.77	35.23
FGS	62.94	37.06	66.03	33.97	62.62	37.38	65.80	34.20	68.78	31.22
MGS	59.84	40.16	58.13	41.87	59.65	40.35	58.53	41.47	60.88	39.12
TDA	40.28	59.72	39.78	60.22	39.12	60.88	38.42	61.58	36.38	63.62
MWS	61.50	38.50	63.82	36.18	62.25	37.75	60.71	39.29	58.38	41.62
DXS	40.72	59.28	42.65	57.35	42.82	57.18	42.87	57.13	41.27	58.73
MGS	53.53	46.47	53.02	46.98	52.87	47.13	51.45	48.55	54.84	45.16

**Table 3** Contributions of climate and land-use change on future richness changes of terrestrial vertebrates (%)

*Note*: TMB, TDB, and TSC denote the biomes of Tropical and Subtropical Moist Broadleaf Forests, Tropical and Subtropical Dry Broadleaf Forests, and Tropical
 and Subtropical Coniferous Forests. TBM, TCF and BRF represent the biomes of Temperate Broadleaf and Mixed Forests, Temperate Coniferous Forests, and Boreal
 Forests/Taiga. TSG, TGS, and FGS are the biomes of Tropical and Subtropical Grasslands, Savannas, and
 Shrublands, and Flooded Grasslands and Savannas. MGS, TDA, and MWS are the biomes of Montane Grasslands and Shrublands, Tundra, and Mediterranean
 Forests, Woodlands, and Scrub. DXS and MGV are the biomes of Deserts and Xeric Shrublands and Mangroves, respectively. The abbreviation of terrestrial biomes
 also can be referred to SI. Table 2.

25 / 41

## 396 **4. Discussion**

Land-use change is a critical driver of historical change of biodiversity under the 397 398 global climate change (Jung, Rowhani, & Scharlemann, 2019). In this study, we assess the combined effects of future climate and land-use change on terrestrial 399 400 vertebrates and separate the relative contributions of climate change and land-use 401 change at a global scale under SSPs. The results reveal various spatial distribution 402 responses of terrestrial vertebrates across the globe, enabling us to understand the 403 combined effects of climate and land-use change, and identifying the priority for 404 biodiversity conservation.

## 405 **4.1 Relative contributions of climate change and land-use change**

406 Climate change has been demonstrated to be a contributor to reshaping the 407 richness and geographical ranges of terrestrial vertebrates, with a relative contribution 408 of 51.48%. The importance of RF models further reveal that mean annual temperature 409 and mean annual precipitation are the dominant climate factors influencing the richness changes of terrestrial vertebrates (SI. Table 3), which is consistent with 410 411 previous studies that measured the effects of climate change from different 412 perspectives (Garcia et al., 2014; Urban, 2015). The climate-related effects on 413 terrestrial vertebrates are different among terrestrial biomes. For all tropical biomes, 414 the diversity of terrestrial vertebrates is largely influenced by mean annual 415 temperature, followed by mean annual precipitation. This phenomenon may originate 26/41

416 from that the tropical species are systematically more sensitive to climate variations 417 than species at other locations (Deutsch et al., 2008; Freeman & Class Freeman, 2014). 418 Higher spatial heterogeneity of climate change usually means wider environmental 419 tolerance (Bonebrake & Mastrandrea, 2010). Since tropical biomes are characterized 420 by low spatial heterogeneity in temperature, species in the tropics, compared with 421 temperate species, have to move farther along latitude to offset the adverse effect of 422 rising temperature (Colwell, Brehm, Cardelús, Gilman, & Longino, 2008). Although 423 complex topography can alleviate the negative effect caused by the warming climate, 424 the niche of the indigenous montane species would be under threat (Elsen & Tingley, 425 2015). Meanwhile, the increase in the length of the dry season caused by the 426 precipitation variation in tropical areas will directly affect the phenology and duration 427 of bird reproductive activities and the availability of food resources, resulting in 428 nearly one-third of tropical birds suffering population decline (Brawn, Benson, Stager, 429 Sly, & Tarwater, 2017).

Land-use change is also identified the other important determinant for terrestrial vertebrates across the globe under all five SSPs, averagely resulting in 48.52% of global richness changes, which is in line with many existed studies (Jantz et al., 2015; Jetz, Wilcove, & Dobson, 2007; Mantyka-Pringle et al., 2015). This suggests that land-use change also plays a significant role in shifting species ranges and changing richness diversity of terrestrial vertebrates. However, the effect of land-use change on terrestrial vertebrates often shows a superposition or mitigation effect on the effect of 27/41 437 climate change. For instance, Jung et al. (2019) claimed that abrupt land-use change 438 could lower local species and abundance by 4.2% and 2.0%, but this loss could 439 completely recover after ten years with a constant climate condition. Moreover, our analysis shows that the land-use naturalness, in some terrestrial biomes (e.g. BRF), 440 441 contributes more to richness changes by comparison with climate variables, such as 442 the mean annual temperature and the mean annual precipitation. This high 443 land-use-related contribution may largely be associated with agricultural expansion 444 (Dobrovolski, Diniz-Filho, Loyola, & De Marco Júnior, 2011). According to the high 445 level of projected population growth (Gerland et al., 2014) and the dietary transitions 446 to more calories and animal-based foods (Willett et al., 2019), more natural land 447 needs to be converted into agricultural land for satisfying basic food systems, thereby 448 making natural habitats more fragmented and leading to species extinctions (Williams 449 et al., 2020).

## 450 **4.2 Regional differences of species richness changes**

The combined effects of climate and land-use change on terrestrial vertebrates show substantial latitudinal differences with a large decline at low latitudes, this result is consistent with those of previous studies which suggest a sharp biodiversity loss at low latitudes (Chaudhary & Mooers, 2018; Schipper et al., 2019). However, unlike numerous studies focusing on a poleward shift of terrestrial vertebrates in the future (Chen et al., 2011; Hickling et al., 2006; Hill et al., 2011), we warn that the richness 457 changes at low latitudes should be paid much more attention to. The main reason is that low-latitude regions have a considerably number of species and the most 458 459 abundant biological resources on the planet (Gaston, 2000; Jenkins et al., 2013). For 460 instance, the Amazon Basin is home to nearly one-quarter of terrestrial species. 461 Besides that, the low latitudes are subject to some of the locations that most disturbed 462 by anthropogenic activities (Barlow et al., 2018), including land-use change and 463 degradation (Keenan et al., 2015), pollution (Lewis, Silburn, Kookana, & Shaw, 2016) and overexploitation (Ingram et al., 2018). Multiple anthropogenic stressors have 464 465 caused tropical ecosystems more vulnerable (Buisson et al., 2019; Cole, Bhagwat, & Willis, 2014) and transform from species-rich systems to species-poor systems 466 467 (Veldman & Putz, 2011).

468 Furthermore, our result at the country-specific level indicates that the richness 469 loss is mainly concentrated in the countries at the middle-income level which is 470 highly consistent with that of the study by Waldron et al.(2017). Their study finds that 471 biodiversity declines as the gross domestic product (GDP) grows, but the effect of 472 GDP growth is not significant in the poorest countries and can be partly offset by improvements in the quality of national governance. Obviously, countries at different 473 474 income levels have different abilities to cope with the effects of climate change, as 475 well as varied social consciousness and paid willingness for biodiversity conservation (Jacobsen & Hanley, 2009; Turpie, 2003), leading to different magnitudes of 476 477 species-richness changes. Meanwhile, the phenomenon is also closely related to 29 / 41

478 economic activities. Some other studies have shown that high-income countries can
479 shift their pressure on species to low- and middle-income countries through importing
480 of products and services (Holland et al., 2019; Lenzen et al., 2012). These telecoupled
481 activities make the country-specific richness changes more complex.

482

## 4.3 Implications for biodiversity conservation

483 Global biodiversity will be affected by both climate change and land-use change, 484 and climate change is considered the dominant cause of species extinction. How 485 society responds to climate change will seriously affect biodiversity changes, because 486 effective climate change mitigation policies will significantly alleviate the direct effect of climate change on biodiversity (Mantyka-Pringle et al., 2015; Schipper et al., 487 488 2019). Our analysis shows that under the scenario of the highest greenhouse gas emissions, that is, the SSP5 scenario, the diversity of terrestrial vertebrates will 489 490 decline the most in the middle of 21st century, and the relative contribution of climate 491 change is also the highest (51.97%), comparing with the climate-related contribution 492 under SSP4 scenario by 50.92%. This demonstrates that our society must immediately 493 implement sustainable development strategies through transforming energy 494 production and consumption, improving renewable energy technologies, reducing greenhouse gas emissions, and slowing down the rate of climate change. 495

What's more, reasonable land-use planning is equally important for biodiversityconservation, especially at a country-specific level. First, reducing deforestation and

30 / 41

498 agricultural expansion are the most direct ways to conserve species through protecting habitats. A market-based protective payment mechanism, such as REDD+ (Agrawal, 499 500 Nepstad, & Chhatre, 2011; McDermott, Coad, Helfgott, & Schroeder, 2012), can be 501 employed to higher the cost of deforestation for private-sector actors (Lambin et al., 502 2018). Improved agricultural production efficiency (Grassini & Cassman, 2012) and 503 proactive food system changes (Booth et al., 2021; Williams et al., 2020) are also 504 essential approaches to reducing biodiversity threats. Second, the establishment of 505 protected areas and protected area networks for extinct species is an effective tool to 506 relieve pressure caused by land-use change (like infrastructure development). The 507 effectiveness of global protected areas is not optimistic as before it designed because 508 of ignoring the importance of management (Jones et al., 2018). By introducing the 509 protected area networks, merely protected areas in Europe have reached the expected 510 effectiveness and have the potential to resist future climate change (Araújo, Alagador, 511 Cabeza, Nogués-Bravo, & Thuiller, 2011). Third, establishing laws for local species is 512 proven to be beneficial to strictly prohibit overexploitation and illegal trade of 513 endangered species (Mothes al.. 2021). Other channels, including et 514 newly-established economic regulations like payment for ecosystem services (Grima, 515 Singh, Smetschka, & Ringhofer, 2016; Redford & Adams, 2009), are substantial tools 516 as financial supports for biodiversity conservation. In addition, strengthening the 517 cooperation between science and policy at all levels is fundamental to integrate 518 scientific, indigenous and local knowledge to support land-use decision-making at the 31 / 41

519 country-specific level.

## 520 **5. Conclusions**

521 Climate and land-use change are considered major factors causing biodiversity loss. However, previous studies rarely take the combined effect of climate and 522 523 land-use change on global biodiversity. By using empirical data, we assess the 524 combined effect of climate and land-use change on species richness of terrestrial 525 vertebrates and evaluate the relative contributions for climate change and land-use 526 change. Land-use change is evaluated to account for nearly half of future richness 527 changes of global terrestrial vertebrates, but slightly lower than the contribution of climate change. With the combined effect of climate and land-use change, 528 approximately 45.99% of Earth's land would experience richness losses of terrestrial 529 vertebrates, especially in Southeast Asia, sub-Saharan Africa and Latin America. The 530 531 analysis on the country-specific level also shows that nearly half of the countries in 532 the world would confront biodiversity loss, of which 19.62% had average species 533 richness loss rates of over 15%. These findings demonstrate that land-use change, like 534 climate change, plays a comparably significant role in the richness changes of 535 terrestrial vertebrates. More importantly, such insight into attribution analysis of 536 biodiversity loss is required for future biodiversity conservation, such as the Aichi biodiversity targets and the post-2020 global biodiversity framework. 537

538

There are several limitations in our analysis. First, the interspecies relationships

539 and energy requirements are not considered in our species distribution models. Second, the contribution of land-use intensity in our analysis may be underestimated 540 541 as the land-use intensity is calculated by using population density and the naturalness 542 of each land-use class. Although the population density can represent the number of 543 people dwelling in each grid cell, this population aggregation is unable to 544 comprehensively be illustrated by the land-use intensity. Although some limitations to 545 our projections, this paper goes much beyond previous analysis in three main ways: (1) 546 we generate the proxies of land-use naturalness and land-use intensity to quantify the 547 effects of land-use change. (2) we build the relationship between the combined effects 548 of climate and land-use change and diversity changes using machine learning techniques at a global scale. (3) The relative contributions of climate change and 549 550 land-use change to terrestrial vertebrates are firstly assessed, which is critical to 551 mitigation policies and conservation strategies, such as the new post-2020 global 552 biodiversity framework.

## 553 Acknowledgement

This study was supported by the National Key R&D Program of China (Grant No. 2017YFA0604402) and the Key National Natural Science Foundation of China (Grant No. 41531176). The geographical range maps of amphibians and mammals are acquired from IUCN Red List (https://www.iucnredlist.org/), and range maps for birds from the Birdlife International (http://www.birdlife.org/). The historical climate

559	dataset, includii	ng temperature, precipi	tation and wind	d speed, are from	n the Global
560	Surface	Summary	of	the	Day
561	( <u>https://data.noa</u>	a.gov/dataset/global-sur	<u>face-summary-c</u>	of-the-day-gsod).	The
562	elevation grid	that support the find	lings of this	study is availa	ble at FAO
563	(http://www.fao	<u>.org/</u> ). The future climat	e dataset (tempe	erature, precipitat	ion and wind
564	speed in 205	50) adopted in this	study are	derived from	the CMIP6
565	(https://esgf-noc	<u>le.llnl.gov/</u> ). The histor	ical and future	a land-use map	are from the
566	European Space	ce Agency Climate	Change Initiat	ive Land Cove	er (CCI-LC,
567	http://maps.elie.	ucl.ac.be/CCI/viewer/in	<u>dex.php</u> ) and th	ne Integrated Mo	del to assess
568	the Global Envi	ronment (IMAGE, <u>https</u>	://dataplatform.l	knmi.nl/?q=PBL)	, respectively.
569	The historica	al population dens	sity and N	NPP dataset	are from
570	https://landscan.	ornl.gov/landscan-datas	ets and http://f	iles.ntsg.umt.edu/	. The future
571	population den	usity and NPP datas	et used for	projection are	available at
572	https://dataplatfo	orm.knmi.nl/?q=PBL.			

## 573 **References**

Agrawal, A., Nepstad, D., & Chhatre, A. (2011). Reducing Emissions from
Deforestation and Forest Degradation. *Annual Review of Environment and Resources*,
36(1), 373-396. doi:10.1146/annurev-environ-042009-094508

Araújo, M. B., Alagador, D., Cabeza, M., Nogués-Bravo, D., & Thuiller, W.
(2011). Climate change threatens European conservation areas. *Ecology Letters*, 14(5),
484-492. doi:10.1111/j.1461-0248.2011.01610.x

Balvanera, P., Pfisterer, A. B., Buchmann, N., He, J.-S., Nakashizuka, T.,
Raffaelli, D., & Schmid, B. (2006). Quantifying the evidence for biodiversity effects
on ecosystem functioning and services. *Ecology Letters*, 9(10), 1146-1156.
doi:10.1111/j.1461-0248.2006.00963.x

Barbet-Massin, M., Thuiller, W., & Jiguet, F. (2012). The fate of European 584 585 breeding birds under climate, land-use and dispersal scenarios. global Chang Biology, 586 18(3), 881-890. doi:10.1111/j.1365-2486.2011.02552.x 587 Barlow, J., França, F., Gardner, T. A., Hicks, C. C., Lennox, G. D., Berenguer, E., . . . Graham, N. A. J. (2018). The future of hyperdiverse tropical ecosystems. 588 589 Nature, 559(7715), 517-526. doi:10.1038/s41586-018-0301-1 590 BirdLife International NatureServe. (2017). Birdlife data zone. Available at 591 http://www.birdlife.org/. 592 Bonebrake, T. C., & Mastrandrea, M. D. (2010). Tolerance adaptation and 593 precipitation changes complicate latitudinal patterns of climate change impacts. 594 Proceedings of the National Academy of Sciences, 107(28), 12581-12586. 595 doi:10.1073/pnas.0911841107 596 Booth, H., Clark, M., Milner-Gulland, E. J., Amponsah-Mensah, K., Antunes, A. P., Brittain, S., . . . Williams, D. R. (2021). Investigating the risks of removing wild 597

598meatfromglobalfoodsystems.CurrentBiology.599doi:<a href="https://doi.org/10.1016/j.cub.2021.01.079">https://doi.org/10.1016/j.cub.2021.01.079</a>

Brawn, J. D., Benson, T. J., Stager, M., Sly, N. D., & Tarwater, C. E. (2017).
Impacts of changing rainfall regime on the demography of tropical birds. *Nature Climate Change*, 7(2), 133-136. doi:10.1038/nclimate3183

Buisson, E., Le Stradic, S., Silveira, F. A. O., Durigan, G., Overbeck, G. E.,
Fidelis, A., . . . Veldman, J. W. (2019). Resilience and restoration of tropical and
subtropical grasslands, savannas, and grassy woodlands. *Biological Reviews*, 94(2),
590-609. doi:10.1111/brv.12470

607 Chaudhary, A., & Mooers, A. O. (2018). Terrestrial vertebrate biodiversity loss
608 under future global land use change scenarios. *Sustainability*, *10*(8), 2764.

Chen, I.-C., Hill, J. K., Ohlemüller, R., Roy, D. B., & Thomas, C. D. (2011).
Rapid Range Shifts of Species Associated with High Levels of Climate Warming. *Science*, 333(6045), 1024-1026. doi:10.1126/science.1206432, Science

Cole, L. E. S., Bhagwat, S. A., & Willis, K. J. (2014). Recovery and resilience of
tropical forests after disturbance. *Nature Communications*, 5(1), 3906.
doi:10.1038/ncomms4906

Colwell, R. K., Brehm, G., Cardelús, C. L., Gilman, A. C., & Longino, J. T.
(2008). Global Warming, Elevational Range Shifts, and Lowland Biotic Attrition in
the Wet Tropics. *Science*, *322*(5899), 258-261. doi:10.1126/science.1162547

Davies, J., Poulsen, L., Schulte-Herbrüggen, B., Mackinnon, K., Crawhall, N.,
Henwood, W., ... Gudka, M. (2012). *Conserving Dryland Biodiversity. xii+ 84p*.

Deutsch, C. A., Tewksbury, J. J., Huey, R. B., Sheldon, K. S., Ghalambor, C. K.,
Haak, D. C., & Martin, P. R. (2008). Impacts of climate warming on terrestrial
ectotherms across latitude. *Proceedings of the National Academy of Sciences*, *105*(18),
6668-6672. doi:10.1073/pnas.0709472105

624 Di Marco, M., Harwood, T. D., Hoskins, A. J., Ware, C., Hill, S. L. L., & Ferrier,

S. (2019). Projecting impacts of global climate and land-use scenarios on plant
biodiversity using compositional-turnover modelling. *Global change biology*, 25(8),
2763-2778. doi:10.1111/gcb.14663

Dobrovolski, R., Diniz-Filho, J. A. F., Loyola, R. D., & De Marco Júnior, P.
(2011). Agricultural expansion and the fate of global conservation priorities. *Biodiversity and Conservation*, 20(11), 2445-2459. doi:10.1007/s10531-011-9997-z

Dornelas, M., Gotelli, N. J., McGill, B., Shimadzu, H., Moyes, F., Sievers, C., &
Magurran, A. E. (2014). Assemblage Time Series Reveal Biodiversity Change but Not
Systematic Loss. *Science*, *344*(6181), 296-299. doi:10.1126/science.1248484, Science

Dullinger, S., Gattringer, A., Thuiller, W., Moser, D., Zimmermann, N. E.,
Guisan, A., . . . Hülber, K. (2012). Extinction debt of high-mountain plants under
twenty-first-century climate change. *Nature Climate Change*, 2(8), 619-622.
doi:10.1038/nclimate1514

Elsen, P. R., & Tingley, M. W. (2015). Global mountain topography and the fate
of montane species under climate change. *Nature Climate Change*, 5(8), 772-776.
doi:10.1038/nclimate2656

Fadrique, B., Báez, S., Duque, Á., Malizia, A., Blundo, C., Carilla, J., . . . Feeley,
K. J. (2018). Widespread but heterogeneous responses of Andean forests to climate
change. *Nature*, 564(7735), 207-212. doi:10.1038/s41586-018-0715-9

Freeman, B. G., & Class Freeman, A. M. (2014). Rapid upslope shifts in New
Guinean birds illustrate strong distributional responses of tropical montane species to
global warming. *Proceedings of the National Academy of Sciences, 111*(12),
4490-4494. doi:10.1073/pnas.131819011

Garcia, R. A., Cabeza, M., Rahbek, C., & Araújo, M. B. (2014). Multiple
Dimensions of Climate Change and Their Implications for Biodiversity. *Science*, *344*(6183), 1247579. doi:10.1126/science.1247579

Gaston, K. J. (2000). Global patterns in biodiversity. *Nature*, 405(6783), 220-227.
doi:10.1038/35012228

Gerland, P., Raftery, A. E., Ševčíková, H., Li, N., Gu, D., Spoorenberg, T., . . .
Wilmoth, J. (2014). World population stabilization unlikely this century. *Science*, 346(6206), 234-237. doi:10.1126/science.1257469

Grassini, P., & Cassman, K. G. (2012). High-yield maize with large net energy
yield and small global warming intensity. *Proceedings of the National Academy of Sciences*, 109(4), 1074-1079. doi:10.1073/pnas.1116364109

Grima, N., Singh, S. J., Smetschka, B., & Ringhofer, L. (2016). Payment for
Ecosystem Services (PES) in Latin America: Analysing the performance of 40 case
studies. *Ecosystem Services*, 17, 24-32.
doi:https://doi.org/10.1016/j.ecoser.2015.11.010

Hickling, R., Roy, D. B., Hill, J., Fox, R., & Thomas, C. (2006). The
distributions of a wide range of taxonomic groups are expanding polewards. *Global change biology*, *12*. doi:10.1111/j.1365-2486.2006.01116.x

Hill, J. K., Griffiths, H. M., & Thomas, C. D. (2011). Climate Change and
Evolutionary Adaptations at Species' Range Margins. *Annual Review of Entomology*,
56(1), 143-159. doi:10.1146/annurev-ento-120709-144746

Holland, R. A., Scott, K., Agnolucci, P., Rapti, C., Eigenbrod, F., & Taylor, G.
(2019). The influence of the global electric power system on terrestrial biodiversity. *Proceedings of the National Academy of Sciences, 116*(51), 26078-26084.
doi:10.1073/pnas.1909269116

Ingram, D. J., Coad, L., Abernethy, K. A., Maisels, F., Stokes, E. J., Bobo, K.
S., ... Scharlemann, J. P. W. (2018). Assessing Africa-Wide Pangolin Exploitation by
Scaling Local Data. *CONSERVATION LETTERS*, *11*(2), e12389.
doi:https://doi.org/10.1111/conl.12389

Jacobsen, J. B., & Hanley, N. (2009). Are There Income Effects on Global
Willingness to Pay for Biodiversity Conservation? *Environmental and Resource Economics*, 43(2), 137-160. doi:10.1007/s10640-008-9226-8

Jantz, S. M., Barker, B., Brooks, T. M., Chini, L. P., Huang, Q., Moore, R. M., ...
Hurtt, G. C. (2015). Future habitat loss and extinctions driven by land-use change in
biodiversity hotspots under four scenarios of climate-change mitigation. *Conservation Biology*, 29(4), 1122-1131. doi:<u>https://doi.org/10.1111/cobi.12549</u>

Jenkins, C. N., Pimm, S. L., & Joppa, L. N. (2013). Global patterns of terrestrial
vertebrate diversity and conservation. *Proceedings of the National Academy of Sciences*, *110*(28), E2602-E2610. doi:10.1073/pnas.1302251110

Jetz, W., Wilcove, D. S., & Dobson, A. P. (2007). Projected impacts of climate
and land-use change on the global diversity of birds. *PLoS biology*, 5(6), e157-e157.
doi:10.1371/journal.pbio.0050157

Johnson, C. N., Balmford, A., Brook, B. W., Buettel, J. C., Galetti, M.,
Guangchun, L., & Wilmshurst, J. M. (2017). Biodiversity losses and conservation
responses in the Anthropocene. *Science*, *356*(6335), 270-275.
doi:10.1126/science.aam9317

Jones, K. R., Venter, O., Fuller, R. A., Allan, J. R., Maxwell, S. L., Negret, P. J.,
& Watson, J. E. M. (2018). One-third of global protected land is under intense human
pressure. *Science*, *360*(6390), 788-791. doi:10.1126/science.aap9565

Jung, M., Rowhani, P., & Scharlemann, J. P. W. (2019). Impacts of past abrupt
land change on local biodiversity globally. *Nature Communications*, 10(1), 5474.
doi:10.1038/s41467-019-13452-3

Keenan, R. J., Reams, G. A., Achard, F., de Freitas, J. V., Grainger, A., &
Lindquist, E. (2015). Dynamics of global forest area: Results from the FAO Global
Forest Resources Assessment 2015. *Forest Ecology and Management, 352*, 9-20.
doi:https://doi.org/10.1016/j.foreco.2015.06.014

Lambin, E. F., Gibbs, H. K., Heilmayr, R., Carlson, K. M., Fleck, L. C., Garrett,
R. D., . . . Walker, N. F. (2018). The role of supply-chain initiatives in reducing
deforestation. *Nature Climate Change*, 8(2), 109-116.

707 doi:10.1038/s41558-017-0061-1

Lenzen, M., Moran, D., Kanemoto, K., Foran, B., Lobefaro, L., & Geschke, A.
(2012). International trade drives biodiversity threats in developing nations. *Nature*,
486(7401), 109-112. doi:10.1038/nature11145

Lewis, S. E., Silburn, D. M., Kookana, R. S., & Shaw, M. (2016). Pesticide
Behavior, Fate, and Effects in the Tropics: An Overview of the Current State of
Knowledge. *Journal of Agricultural and Food Chemistry*, 64(20), 3917-3924.
doi:10.1021/acs.jafc.6b01320

Liu, X., Pei, F., Wen, Y., Li, X., Wang, S., Wu, C., . . . Liu, Z. (2019). Global
urban expansion offsets climate-driven increases in terrestrial net primary productivity. *Nature Communications*, *10*(1), 5558. doi:10.1038/s41467-019-13462-1

Maiorano, L., Falcucci, A., Garton, E., & BOITANI, L. (2007). Contribution of
the Natura 2000 Network to Biodiversity Conservation in Italy. *Society of conservation biology*, 21(6), 1433-1444. doi:10.1111/j.1523-1739.2007.00831.x

Manaye, A., Negash, M., & Alebachew, M. (2019). Effect of degraded land
rehabilitation on carbon stocks and biodiversity in semi-arid region of Northern
Ethiopia. *Forest Science and Technology*, 15(2), 70-79.
doi:10.1080/21580103.2019.1592787

Mantyka-pringle, C. S., Martin, T. G., & Rhodes, J. R. (2012). Interactions
between climate and habitat loss effects on biodiversity: a systematic review and
meta-analysis. *Global change biology*, *18*(4), 1239-1252.
doi:10.1111/j.1365-2486.2011.02593.x

Mantyka-Pringle, C. S., Visconti, P., Di Marco, M., Martin, T. G., Rondinini, C.,
& Rhodes, J. R. (2015). Climate change modifies risk of global biodiversity loss due
to land-cover change. *Biological Conservation*, 187, 103-111.
doi:<u>https://doi.org/10.1016/j.biocon.2015.04.016</u>

Marmion, M., Parviainen, M., Luoto, M., Heikkinen, R. K., & Thuiller, W.
(2009). Evaluation of consensus methods in predictive species distribution modelling. *Diversity and Distributions*, 15(1), 59-69.

Maxwell, S. L., Fuller, R. A., Brooks, T. M., & Watson, J. E. (2016). Biodiversity:
The ravages of guns, nets and bulldozers. *Nature News*, *536*(7615), 143.

738 McDermott, C. L., Coad, L., Helfgott, A., & Schroeder, H. (2012). 739 social safeguards in REDD+: actors, Operationalizing interests and ideas. 740 Environmental Science & Policy, 21, 63-72. 741 doi:https://doi.org/10.1016/j.envsci.2012.02.007

Midgley, G. F. (2012). Biodiversity and Ecosystem Function. *Science*, *335*(6065),
174-175. doi:10.1126/science.1217245 %J Science

Montesino Pouzols, F., Toivonen, T., Di Minin, E., Kukkala, A. S., Kullberg, P.,
Kuusterä, J., . . Moilanen, A. (2014). Global protected area expansion is
compromised by projected land-use and parochialism. *Nature*, *516*(7531), 383-386.
doi:10.1038/nature14032

Mothes, C. C., Stemle, L. R., Fonseca, T. N., Clements, S. L., Howell, H. J., &
Searcy, C. A. (2021). Protect or perish: Quantitative analysis of state-level species
protection supports preservation of the Endangered Species Act. *CONSERVATION LETTERS*, 14(1), e12761. doi:<u>https://doi.org/10.1111/conl.12761</u>

Newbold, T., Hudson, L. N., Arnell, A. P., Contu, S., De Palma, A., Ferrier, S., ...
Purvis, A. (2016). Has land use pushed terrestrial biodiversity beyond the planetary
boundary? A global assessment. *Science*, *353*(6296), 288-291.
doi:10.1126/science.aaf2201

Olson, D. M., Dinerstein, E., Wikramanayake, E. D., Burgess, N. D., Powell, G.
V. N., Underwood, E. C., . . . Kassem, K. R. (2001). Terrestrial Ecoregions of the
World: A New Map of Life on Earth: A new global map of terrestrial ecoregions
provides an innovative tool for conserving biodiversity. *BioScience*, *51*(11), 933-938.
doi:10.1641/0006-3568(2001)051[0933:TEOTWA]2.0.CO;2

Parmesan, C. (2006). Ecological and Evolutionary Responses to Recent Climate
Change. *Review in Advance*, 37(1), 637-669.
doi:10.1146/annurev.ecolsys.37.091305.110100

Pauli, H., Gottfried, M., Dullinger, S., Abdaladze, O., Akhalkatsi, M., Alonso, J.
L. B., . . . Grabherr, G. (2012). Recent Plant Diversity Changes on Europe's Mountain
Summits. *Science*, *336*(6079), 353-355. doi:10.1126/science.1219033 %J Science

Pekin, B. K., & Pijanowski, B. C. (2012). Global land use intensity and the
endangerment status of mammal species. *Diversity and Distributions*, 18(9), 909-918.
doi:10.1111/j.1472-4642.2012.00928.x

Peters, M. K., Hemp, A., Appelhans, T., Becker, J. N., Behler, C., Classen, A., ...
Steffan-Dewenter, I. (2019). Climate–land-use interactions shape tropical mountain
biodiversity and ecosystem functions. *Nature*, 568(7750), 88-92.
doi:10.1038/s41586-019-1048-z

Popp, A., Calvin, K., Fujimori, S., Havlik, P., Humpenöder, F., Stehfest, E., . . .
Vuuren, D. P. v. (2017). Land-use futures in the shared socio-economic pathways. *Global Environmental Change*, 42, 331-345.
doi:https://doi.org/10.1016/j.gloenvcha.2016.10.002

Porter, W. P., Budaraju, S., Stewart, W. E., & Ramankutty, N. (2015). Calculating
Climate Effects on Birds and Mammals: Impacts on Biodiversity, Conservation,
Population Parameters, and Global Community Structure1. *American Zoologist*, 40(4),
597-630. doi:10.1093/icb/40.4.597

Redford, K. H., & Adams, W. M. (2009). Payment for Ecosystem Services and
the Challenge of Saving Nature. *Conservation Biology*, 23(4), 785-787.
doi:<u>https://doi.org/10.1111/j.1523-1739.2009.01271.x</u>

Riahi, K., van Vuuren, D. P., Kriegler, E., Edmonds, J., O'Neill, B. C., Fujimori,
S., ... Tavoni, M. (2017). The Shared Socioeconomic Pathways and their energy, land
use, and greenhouse gas emissions implications: An overview. *Global Environmental Change*, 42, 153-168. doi:<u>https://doi.org/10.1016/j.gloenvcha.2016.05.009</u>

789 Rojas, C., Munizaga, J., Rojas, O., Martínez, C., & Pino, J. (2019). Urban 790 development versus wetland loss in a coastal Latin American city: Lessons for 791 sustainable planning. Land Use land use Policy, 80, 47-56. 792 doi:https://doi.org/10.1016/j.landusepol.2018.09.036

793 Schipper, A. M., Hilbers, J. P., Meijer, J. R., Antão, L. H., Benítez-López, A., de 794 Jonge, M. M. J., . . . Huijbregts, M. A. J. (2019). Projecting terrestrial biodiversity 795 Chang intactness with GLOBIO 4. global Biology, 26(2),760-771. 796 doi:10.1111/gcb.14848

797 Shi, H., Li, X., Liu, X., Wang, S., Liu, X., Zhang, H., . . . Li, T. (2020). Global 798 protected areas boost the carbon sequestration capacity: Evidences from econometric 799 causal analysis. Science of The Total Environment, 715, 137001. 800 doi:https://doi.org/10.1016/j.scitotenv.2020.137001

Thuiller, W., Lafourcade, B., Engler, R., & Araújo, M. B. (2009). BIOMOD – a
platform for ensemble forecasting of species distributions. *Ecography*, 32(3), 369-373.
doi:10.1111/j.1600-0587.2008.05742.x

Turpie, J. K. (2003). The existence value of biodiversity in South Africa: how
interest, experience, knowledge, income and perceived level of threat influence local
willingness to pay. *Ecological Economics*, 46(2), 199-216.
doi:https://doi.org/10.1016/S0921-8009(03)00122-8

808 Urban, M. C. (2015). Accelerating extinction risk from climate change. *Science*,
809 348(6234), 571-573. doi:10.1126/science.aaa4984

Veldman, J. W., & Putz, F. E. (2011). Grass-dominated vegetation, not
species-diverse natural savanna, replaces degraded tropical forests on the southern
edge of the Amazon Basin. *Biological Conservation*, 144(5), 1419-1429.
doi:https://doi.org/10.1016/j.biocon.2011.01.011

Waldron, A., Miller, D. C., Redding, D., Mooers, A., Kuhn, T. S., Nibbelink,
N., . . . Gittleman, J. L. (2017). Reductions in global biodiversity loss predicted from
conservation spending. *Nature*, 551(7680), 364-367. doi:10.1038/nature24295

Wardle, D. A., Bardgett, R. D., Callaway, R. M., & Van der Putten, W. H. (2011).
Terrestrial Ecosystem Responses to Species Gains and Losses. *Science*, *332*(6035),
1273-1277. doi:10.1126/science.1197479 %J Science

820 Willett, W., Rockström, J., Loken, B., Springmann, M., Lang, T., Vermeulen, 821 S., ... Murray, C. (2019). Food in the Anthropocene: the EAT-Lancet Commission on 822 healthy diets from sustainable food systems. The Lancet, 393. 823 doi:10.1016/S0140-6736(18)31788-4

Williams, D. R., Clark, M., Buchanan, G. M., Ficetola, G. F., Rondinini, C., &
Tilman, D. (2020). Proactive conservation to prevent habitat losses to agricultural
expansion. *Nature Sustainability*. doi:10.1038/s41893-020-00656-5

Wu, S., Zhou, S., Chen, D., Wei, Z., Dai, L., & Li, X. (2014). Determining the
contributions of urbanisation and climate change to NPP variations over the last
decade in the Yangtze River Delta, China. *Science of The Total Environment*, 472,

- 830 397-406. doi:<u>https://doi.org/10.1016/j.scitotenv.2013.10.128</u>
- 831 Xu, H., Cao, Y., Yu, D., Cao, M., He, Y., Gill, M., & Pereira, H. M. (2021).
- 832 Ensuring effective implementation of the post-2020 global biodiversity targets. *Nature*
- 833 *Ecology & Evolution*. doi:10.1038/s41559-020-01375-y

Figure 1.





Figure 2.



a The rate of species richness changes

**b** The number of species richness changes



Figure 3.



Figure 4.



The average of net richness loss (%) at the country level (top 50 countries)

Land categories (IPCC)	Average naturalness	Land categories (CCI-LC)	Naturalness
Agriculture	0.22	Rained cropland	0.20
		Irrigated cropland	0.25
		Mosaic cropland (>50%) / natural vegetation (tree, shrub, herbaceous cover) (<50%)	0.30
		Mosaic natural vegetation (tree, shrub, herbaceous cover)	0.90
		(>50%) / cropland (< 50%)	
Forest	0.87	Tree cover, broadleaved, evergreen, closed to open (>15%)	0.95
		Tree cover, broadleaved, deciduous, closed to open (> 15%)	0.90
		Tree cover, needleleaved, evergreen, closed to open (> 15%) $\rangle$	0.90
		Tree cover, needleleaved, deciduous, closed to open (> 15%)	0.85
		Tree cover, mixed leaf type (broadleaved and needleleaved)	0.70
		Mosaic tree and shrub (>50%) / herbaceous cover (< 50%)	0.60
		Tree cover, flooded, fresh or brakish water	0.50
		Tree cover, flooded, saline water	0.45
Grassland	0.77	Mosaic herbaceous cover (>50%) / tree and shrub (<50%)	0.40
		Grassland	0.80
Wetland	0.85	Shrub or herbaceous cover, flooded, fresh-saline or brakish	0.85
		water	
Settlement	0.00	Urban	0.00
Other	0.17	Shrubland	0.30
		Lichens and mosses	0.15
		Sparse vegetation (tree, shrub, herbaceous cover)	0.20
		Bare areas	0.10

**Table 1** The correspondence between the land categories and values of naturalness in each land classes

SSPs	<b>Terrestrial vertebrates</b>	Amphibians	Mammals	Birds
SSP1	45.98	40.90	42.94	38.92
SSP2	46.00	40.52	42.62	38.94
SSP3	45.82	40.75	42.82	38.61
SSP4	45.85	40.21	42.37	38.75
SSP5	46.29	41.13	43.07	39.16

Table 2 The percentages of terrestrial land with richness loss under SSPs (%)

	SSP1		SS	SP 2	SS	SP 3	SS	SP 4	SSP 5		
Biomes	climate	land-use									
	change	change									
TMB	45.32	54.68	46.50	53.50	46.46	53.54	45.91	54.09	46.33	53.67	
TDB	46.86	53.14	47.44	52.56	49.35	50.65	47.14	52.86	51.10	48.90	
TSC	60.42	39.58	64.01	35.99	63.78	36.22	61.15	38.85	66.05	33.95	
TBM	48.40	51.60	47.31	52.69	48.10	51.90	45.59	54.41	47.95	52.05	
TCF	54.25	45.75	51.60	48.40	53.30	46.70	52.88	47.12	52.19	47.81	
BRF	31.44	68.56	30.37	69.63	30.57	69.43	30.65	69.35	31.28	68.72	
TSG	45.84	54.16	47.04	52.96	47.64	52.36	46.36	53.64	47.36	52.64	
TGS	65.98	34.02	63.99	36.01	65.72	34.28	65.45	34.55	64.77	35.23	
FGS	62.94	37.06	66.03	33.97	62.62	37.38	65.80	34.20	68.78	31.22	
MGS	59.84	40.16	58.13	41.87	59.65	40.35	58.53	41.47	60.88	39.12	
TDA	40.28	59.72	39.78	60.22	39.12	60.88	38.42	61.58	36.38	63.62	
MWS	61.50	38.50	63.82	36.18	62.25	37.75	60.71	39.29	58.38	41.62	
DXS	40.72	59.28	42.65	57.35	42.82	57.18	42.87	57.13	41.27	58.73	
MGS	53.53	46.47	53.02	46.98	52.87	47.13	51.45	48.55	54.84	45.16	

Table 3 Contributions of climate and land-use change on future richness changes of terrestrial vertebrates (%)

Note: TMB, TDB, and TSC denote the biomes of Tropical and Subtropical Moist Broadleaf Forests, Tropical and Subtropical Dry Broadleaf Forests, and Tropical and Subtropical Coniferous Forests. TBM, TCF and BRF represent the biomes of Temperate Broadleaf and Mixed Forests, Temperate Coniferous Forests, and Boreal Forests/Taiga. TSG, TGS, and FGS are the biomes of Tropical and Subtropical Grasslands, Savannas, and Shrublands, Temperate Grasslands, Savannas, and Shrublands, and Flooded Grasslands and Savannas. MGS, TDA, and MWS are the biomes of Montane Grasslands and Shrublands, Tundra, and Mediterranean Forests, Woodlands, and Scrub. DXS and MGV are the biomes of Deserts and Xeric Shrublands and Mangroves, respectively. The abbreviation of terrestrial biomes also can be referred to SI. Table 2.

1	L	and-use change contributing almost half of future diversity change
2		of global terrestrial vertebrates under climate change
3	Xia	aojuan Liu <sup>a</sup> , Xia Li <sup>a*</sup> , Jinbao Zhang <sup>b</sup> , Han Zhang <sup>b</sup> , Hong Shi <sup>c</sup> , Yuchao Yan <sup>d</sup> , Han Zhang <sup>a</sup>
4	a.	Key Lab. of Geographic Information Science (Ministry of Education), School of Geographic
5		Sciences, East China Normal University, Shanghai 200241, P.R. China
6	b.	School of Geography and Planning, Guangdong Key Laboratory for Urbanization and
7		Geo-simulation, Sun Yat-sen University, Guangzhou 510275, China.
8	c.	School of Tourism and Historical Culture, Southwest Minzu university, Chengdu 610064, P.R.
9		China
10	d.	Sino-French Institute for Earth System Science, College of Urban and Environmental Sciences,
11		Peking University, Beijing 100871, P.R. China
12	*C	orresponding authors: Xia Li (lixia@geo.ecnu.edu.cn)
13		Key Lab of Geographic Information Science (Ministry of Education),
14		School of Geographic Sciences,
15		East China Normal University,

16 500 Dongchuan Rd., Shanghai 200241, P.R. China.

## 17 Supporting Information (SI)



19 SI. Figure 1 The visualization of a Pearson correlation matrix of all climate and land-use predictors and species richness. Lower, Pearson correlation coefficients. 20 Upper, the significant levels of each Pearson correlation coefficient, \*\*\* means the 21 level of significance p<0.001, \*\* and \* indicate p<0.05 and p<0.1, respectively. ELE, 22 MAP, MAT, MWS are mean elevation, mean annual precipitation, mean annual 23 temperature and mean wind speed. LUN and LUI denote the land-use naturalness and 24 25 land-use intensity. SR, amphibians, mammals and birds are species richness of vertebrate, amphibians, mammals and birds. 26



27

SI. Figure 2 Annual changes of species richness in elevation between 2017 and 2050
 under SSP3 for a terrestrial vertebrates, b amphibians, c mammals and d birds. Colour
 bar shows the number of grid cells that located in annual change of richness species
 and latitude.



**SI. Figure 3** The number of richness change of terrestrial vertebrates from the present to 2050 under SSPs. **a**: SSP1, **b**: SSP2, **c**: SSP3, **d**: SSP4, **e**: SSP5.



SI. Figure 4 The number of richness change of amphibians from the present to 2050 under SSPs. a: SSP1, b: SSP2, c: SSP3, d: SSP4, e: SSP5.



SI. Figure 5 The number of richness change of mammals from the present to 2050 under SSPs. a: SSP1, b: SSP2, c: SSP3, d: SSP4, e: SSP5.



SI. Figure 6 The number of richness change of birds from the present to 2050 under SSPs. a: SSP1, b: SSP2, c: SSP3, d: SSP4, e: SSP5.

		Terrestria	al vertebrates	Am	phibians	Ma	ammals	Birds		
Biomes	Method	Only	With climate	Only	With climate	Only	With climate	Only	With climate	
		climate	and land-use	climate	and land-use	climate	and land-use	climate	and land-use	
		variables	variables	variables	variables	variables	variables	variables	variables	
TMB	RF	0.52	0.56	0.64	0.68	0.40	0.51	0.47	0.51	
	GAM	0.36	0.41	0.49	0.52	0.29	0.34	0.27	0.30	
TDB	RF	0.67	0.68	0.73	0.73	0.70	0.69	0.58	0.59	
	GAM	0.48	0.51	0.51	0.54	0.51	0.52	0.32	0.34	
TSC	RF	0.78	0.79	0.70	0.70	0.81	0.81	0.70	0.72	
	GAM	0.54	0.55	0.47	0.49	0.51	0.53	0.56	0.59	
TBM	RF	0.56	0.60	0.60	0.63	0.51	0.51	0.48	0.49	
	GAM	0.36	0.44	0.51	0.54	0.35	0.40	0.35	0.40	
TCF	RF	0.71	0.73	0.90	0.91	0.65	0.66	0.78	0.80	
	GAM	0.50	0.57	0.82	0.83	0.43	0.48	0.62	0.65	
BRF	RF	0.63	0.67	0.54	0.60	0.65	0.62	0.61	0.66	
	GAM	0.53	0.62	0.33	0.41	0.58	0.63	0.46	0.57	
TSG	RF	0.58	0.66	0.62	0.70	0.54	0.64	0.55	0.58	
	GAM	0.39	0.57	0.46	0.61	0.40	0.58	0.28	0.41	
TGS	RF	0.78	0.74	0.77	0.69	0.76	0.76	0.68	0.68	
	GAM	0.51	0.61	0.54	0.61	0.60	0.64	0.53	0.62	
FGS	RF	0.96	0.96	0.96	0.96	0.96	0.94	0.94	0.93	
	GAM	0.85	0.88	0.87	0.89	0.82	0.84	0.83	0.85	
MGS	RF	0.80	0.84	0.84	0.84	0.76	0.81	0.79	0.81	

	GAM	0.63	0.72	0.59	0.66	0.56	0.66	0.65	0.72
TDA	RF	0.75	0.85	0.66	0.70	0.74	0.83	0.73	0.81
	GAM	0.63	0.81	0.53	0.62	0.64	0.80	0.54	0.72
MWS	RF	0.54	0.56	0.45	0.49	0.75	0.77	0.79	0.80
	GAM	0.32	0.38	0.26	0.33	0.53	0.56	0.61	0.63
DXS	RF	0.65	0.79	0.69	0.78	0.68	0.79	0.64	0.78
	GAM	0.46	0.65	0.43	0.56	0.44	0.60	0.50	0.71
MGV	RF	0.41	0.66	0.76	0.76	0.59	0.62	0.64	0.68
	GAM	0.22	0.43	0.39	0.41	0.34	0.37	0.34	0.38

50 Note: The abbreviation of terrestrial biomes can be referred to SI. Table 2.

	and the corresponding addreviations								
Abbreviations	Terrestrial biomes								
TMB	Tropical and Subtropical Moist Broadleaf Forests								
TDB	Tropical and Subtropical Dry Broadleaf Forests								
TSC	Tropical and Subtropical Coniferous Forests								
TBM	Temperate Broadleaf and Mixed Forests								
TCF	Temperate Coniferous Forests								
BRF	Boreal Forests/Taiga								
TSG	Tropical and Subtropical Grasslands, Savannas, and Shrublands								
TGS	Temperate Grasslands, Savannas, and Shrublands								
FGS	Flooded Grasslands and Savannas								
MGS	Montane Grasslands and Shrublands								
TDA	Tundra								
MWS	Mediterranean Forests, Woodlands, and Scrub								
DXS	Deserts and Xeric Shrublands								
MGV	Mangroves								

**SI. Table 2** The terrestrial biomes from terrestrial ecoregions of the world (TEOW) and the corresponding abbreviations 

Biomes	TMD	TDB	TSC	TBM	TCF	BRF	TSG	TGS	FGS	MGS	TDA	MWS	DXS	MGV
Terrestrial vertebrates														
ELE	0.10	0.02	0.11	0.09	0.15	0.07	0.04	0.06	0.07	0.09	0.49	0.03	0.02	0.02
MAP	0.42	0.30	0.31	0.08	0.11	0.11	0.11	0.01	0.20	0.05	0.02	0.08	0.03	0.25
MAT	0.16	0.45	0.28	0.29	0.55	0.18	0.06	0.23	0.60	0.28	0.04	0.20	0.18	0.12
MWS	0.07	0.08	0.26	0.40	0.08	0.17	0.05	0.24	0.03	0.03	0.07	0.41	0.09	0.12
LUN	0.13	0.10	0.04	0.13	0.10	0.46	0.74	0.44	0.10	0.54	0.38	0.22	0.65	0.39
LUI	0.13	0.04	0.00	0.01	0.02	0.02	0.01	0.02	0.00	0.00	0.00	0.07	0.03	0.10
Amphibi	ans													
ELE	0.14	0.14	0.08	0.13	0.73	0.09	0.03	0.05	0.03	0.12	0.49	0.07	0.04	0.02
MAP	0.55	0.25	0.40	0.25	0.02	0.12	0.26	0.04	0.65	0.11	0.04	0.06	0.02	0.38
MAT	0.11	0.43	0.29	0.31	0.21	0.21	0.09	0.44	0.25	0.50	0.16	0.25	0.23	0.31
MWS	0.05	0.06	0.19	0.23	0.01	0.16	0.06	0.19	0.05	0.01	0.08	0.09	0.07	0.25
LUN	0.04	0.02	0.04	0.07	0.02	0.41	0.57	0.25	0.02	0.25	0.23	0.43	0.46	0.03
LUI	0.11	0.09	0.01	0.01	0.00	0.01	0.00	0.02	0.00	0.00	0.00	0.11	0.18	0.01
Mamma	ls													
ELE	0.0	0.03	0.06	0.11	0.35	0.08	0.08	0.02	0.13	0.08	0.45	0.01	0.04	0.04
MAP	0.21	0.30	0.39	0.08	0.14	0.17	0.09	0.05	0.09	0.06	0.02	0.10	0.03	0.27
MAT	0.22	0.4	0.34	0.30	0.19	0.23	0.06	0.72	0.65	0.27	0.09	0.61	0.23	0.29
MWS	0.07	0.11	0.17	0.45	0.11	0.16	0.04	0.13	0.04	0.04	0.10	0.20	0.09	0.30
LUN	0.14	0.11	0.03	0.06	0.18	0.37	0.71	0.08	0.09	0.54	0.33	0.06	0.53	0.08
LUI	0.27	0.02	0.01	0.01	0.04	0.00	0.03	0.00	0.00	0.01	0.00	0.01	0.09	0.03
Birds														
ELE	0.11	0.07	0.04	0.03	0.07	0.11	0.07	0.20	0.03	0.07	0.49	0.15	0.04	0.07
MAP	0.42	0.37	0.37	0.14	0.15	0.08	0.21	0.04	0.50	0.10	0.04	0.13	0.03	0.33
MAT	0.22	0.30	0.12	0.45	0.69	0.17	0.18	0.22	0.15	0.26	0.08	0.50	0.07	0.21
MWS	0.14	0.18	0.41	0.24	0.04	0.17	0.09	0.08	0.13	0.03	0.01	0.15	0.13	0.23
LUN	0.11	0.05	0.04	0.14	0.05	0.46	0.45	0.45	0.16	0.53	0.39	0.06	0.72	0.12
LUI	0.01	0.03	0.01	0.00	0.00	0.01	0.01	0.01	0.02	0.01	0.00	0.02	0.01	0.04

SI. Table 3 The random forest importance of climate and land-use factors

*Note*: The abbreviation of terrestrial biomes can be referred to **SI Table 2**.